ORIGINAL ARTICLE



Water restriction in viviparous lizards causes transgenerational effects on behavioral anxiety and immediate effects on exploration behavior

David Rozen-Rechels¹ · Andréaz Dupoué¹ · Sandrine Meylan^{1,2} · Beatriz Decencière³ · Sophie Guingand¹ · Jean-François Le Galliard^{1,3}

Received: 20 September 2017 / Revised: 7 January 2018 / Accepted: 11 January 2018 © Springer-Verlag GmbH Germany, part of Springer Nature 2018

Abstract

Behavioral plasticity induced by maternal effects is crucial in adjusting offspring phenotype to match the environment. In particular, changes in water availability during development may initiate a range of behavioral responses, such as natal dispersal, but the contribution of maternal effects from water stress in explaining behavioral variation has been overlooked so far. In this study, we examined behavioral plasticity of juvenile common lizards (*Zootoca vivipara*) in response to changes in pre-natal and post-natal hydric conditions. We exposed pregnant mothers either to a moderate water restriction or to ad libitum access to water. We used neutral arena tests to measure repeatedly activity and space use behaviors of their offspring, either in dry or wet soil conditions, at two distinct life stages (juveniles and yearlings) in order to quantify behavioral flexibility, age differences, and personalities (consistent inter-individual differences). Juveniles and yearlings had greater exploration scores in dry than wet conditions regardless of the pre-natal conditions. Pre-natal water restriction and dry post-natal conditions enhanced thigmotaxis (i.e., a behavior related to anxiety) in juveniles. Maternal environment did not influence behavioral flexibility nor the behavior of yearlings. Behaviors were barely repeatable in juveniles and inconsistent through early life. Assuming that dry conditions may initiate flight responses, these results suggest complex relationships between the developmental timing of water stress, exploration and anxiety behaviors, and dispersal syndromes.

Significance statement

Water restriction should initiate a range of behaviors related to a flight response. However, there is little evidence that changes in water availability across development influence offspring behavior in wild animals. Experiments with lizards demonstrate that maternal water restriction increased the anxiety of their juveniles, whereas their exploration was higher in dry environments. Thus, water availability may be a critical determinant of space use behavior and dispersal syndromes.

Keywords Maternal effects · Exploration · Thigmotaxis · Stress · Reptiles · Water availability

Communicated by T. Madsen

Electronic supplementary material The online version of this article (https://doi.org/10.1007/s00265-018-2443-4) contains supplementary material, which is available to authorized users.

David Rozen-Rechels david.rozen_rechels@upmc.fr

- ¹ Sorbonne Université, CNRS, IRD, INRA, Institut d'Ecologie et des Sciences de l'Environnement de Paris, iEES-Paris, 75252, Paris, France
- ² Sorbonne Université, École Supérieure du Professorat et de l'Éducation de l'Académie de Paris, ESPE Paris, 75016, Paris, France
- ³ École Normale Supérieure, PSL Research University, Département de Biologie, CNRS, UMS 3194, Centre de Recherche en Ecologie Expérimentale et Prédictive (CEREEP-Ecotron IleDeFrance), 78 rue du château, 77140 Saint-Pierre-lès-Nemours, France

Introduction

Water availability and ambient humidity (i.e., hydric conditions) are critical determinants of general ecological patterns as they influence dehydration risk (e.g., Seebacher and Alford 2002; Shochat et al. 2004; Rozen-Rechels et al. 2015; Dupoué et al. 2016, 2017). When water becomes limited, animals may alter a range of behavioral traits including changes in drinking behavior (Rautenberg et al. 1980), changes in posture or decreased locomotor activity to limit water loss (Pough et al. 1983; Lorenzon et al. 1999; Davies and DeNardo 2009; Pintor et al. 2016), and/or shifts in micro-habitat use (Chown et al. 2011; Rozen-Rechels et al. 2015). Dry conditions may also elicit flight responses since limited water availability drives dispersal behaviors (e.g., Massot et al. 2002). Collectively, these behaviors may contribute to regulate the water balance and can influence the fitness of animals in their natural environments (Kearney et al. 2013). Yet, sources of variation in behavioral hydroregulation (i.e., behaviors involved in the regulation of water intake and water loss) among individuals remain poorly investigated so far.

One of the main sources of variation of hydroregulation behaviors, as well as other behaviors, is likely to be intraindividual behavioral flexibility. Behavioral flexibility (also often called behavioral plasticity) is a crucial component of individual responses to changes in environmental conditions that can determine the capacity of organisms to match their environment (Lima and Dill 1990; Dingemanse et al. 2010). In the short term, behavioral flexibility involves fast and reversible changes in response to the environment. Another source of variation of hydration behaviors could be transgenerational effects. Phenotypic traits of mothers and/or environmental conditions encountered by mothers during early development can indeed influence the phenotype of their offspring at birth or later in life (i.e., maternal effects, Mousseau and Fox 1998). Examples of maternal effects include not only adaptive effects that enhance the offspring fitness (e.g., transgenerational induction of defenses against predators, Agrawal et al. 1999) but also non-adaptive responses that decrease the offspring fitness (e.g., overcrowding effects on offspring size, McCormick 2006). Maternal effects can strongly influence mean behavior as well as behavioral flexibility given the influence of maternal environment on fetal neural development (Forstmeier et al. 2004). Yet, maternal effects on offspring behavior induced by water restriction have not been investigated so far. A range of studies in insects and amphibians have shown that these animals are able to adjust flexibly their behavior to control their rates of desiccation and water intake (e.g., Chown et al. 2011; Köhler et al. 2011). In addition, animals exposed to desiccating conditions tend to acclimate both their behavior and their physiology in order to reduce their water loss rates (e.g., Hertz et al. 1979; Chown et al. 2011). Still, whether these functional responses in adults carry over to influence offspring behavior remains an open question.

Another potential source of variation in behavioral responses to water availability is intra-individual change with age, although age-dependent sensitivity of behavior to water availability has still not been investigated. Differences in behavioral flexibility with age could be due to an age-dependent physiological sensitivity to desiccation or may be related to the size scaling of water loss and dehydration tolerance. Evaporative water loss seems to be independent from age in birds (Versteegh et al. 2008); on the contrary, small individuals are more sensitive to water loss than adults in toads (e.g., Newman and Dunham 1994), in insects (Chown and Gaston 1999), and in plants and animals in general (Woods and Smith 2010). Thus, the behavior of juvenile lizards might be more sensitive to changes in their water environment because they are smaller than adults on average. In addition, dispersal and habitat selection abilities vary through life, and behavioral responses to water availability may thus involve either flight or acclimation responses depending on the age of individuals. Because common lizards disperse mainly at the juvenile life stage (e.g., Massot et al. 2002; Meylan et al. 2002), we expect stronger flight responses to water restriction in this age class than in yearlings or more than 2-year-old adults. Recently, it has also been emphasized that individuals of the same age can be highly variable in their behavioral flexibility because individuals vary in their "personalities" (e.g., Dall et al. 2004; Sih et al. 2004; Le Galliard et al. 2013). Personalities imply a low intra-individual and a high inter-individual behavioral flexibility (Dingemanse et al. 2010). Thus, studies of behavioral plasticity should account for both intra-individual and interindividual responses (Dingemanse et al. 2010).

We addressed these three issues (i.e., maternal effects, age dependence, and personality) by examining the effects of maternal water restriction and post-natal habitat humidity on the space use behavior of offspring at two distinct life stages in the European common lizard, Zootoca vivipara (Lichtenstein, 1823), a small Lacertid widely distributed in Eurasia. Populations of this species are dependent on permanent access to free standing water and humid conditions for their longterm persistence (Lorenzon et al. 2001). Limited water availability can impair reproduction and has also direct effects on behavioral activity and natal dispersal behavior; that is, juveniles from dry habitats disperse more, but yearlings are less active when chronically exposed to water restriction (Lorenzon et al. 1999; Massot et al. 2002). In juveniles, behavioral responses to water restriction may thus be explained by their relationship with natal dispersal, whereas behavioral flexibility in yearlings and more than 2-year-old adults may be related to tactics to reduce energy expenditure and water loss or to increase water intake, but not dispersal which is 10 times less on average after the juvenile life stage. However, effects of maternal water restriction on offspring behavior from

juvenile to the yearling stage have not been examined so far. Here, we compared the space use behavior of offspring from mothers exposed to a water restriction period for 2 weeks in mid pregnancy with those from mothers with water ad libitum, hereafter called the "maternal treatment." Behavior was scored 1 day after birth (juveniles) and at the age of 10– 11 months (yearlings) by the use of the neutral arena test (Réale et al. 2007). In order to quantify behavioral flexibility, post-natal conditions were manipulated in the neutral arena test by changing the soil moisture (dry or wet), hereafter called the "soil moisture treatment." Individuals were tested two times in each post-natal condition at each life stage to assess their personalities (Dingemanse and Dochtermann 2013).

We predicted that exposure to a mild water restriction during pregnancy would enhance exploration behavior of offspring (e.g., Greenberg 2002), especially at the juvenile life stage when lizards should be more sensitive to desiccation and are more prone to disperse than yearlings (Newman and Dunham 1994; Massot et al. 2002; Meylan et al. 2002). Lizards should also invest more in exploration behavior in dry post-natal environments (e.g., Lorenzon et al. 1999; Massot et al. 2002), and juveniles from water-restricted mothers should have a stronger behavioral response in dry environments due to an interaction between pre- and postnatal environments. In addition, we expected significant and consistent inter-individual variation in exploration behaviors (Le Galliard et al. 2013) and in behavioral flexibility (Dingemanse et al. 2010).

Material and methods

Data availability statement The datasets generated during and/or analyzed during the current study are available from the corresponding author on reasonable request.

Study species

Common lizard studied individuals were captured from captive populations maintained in 24 outdoor enclosures (10×10 m) in the Centre de Recherche en Écologie Expérimentale et Prédictive (CEREEP), France (48° 17' N, 2° 41' E). Enclosures have a permanent vegetation layer (mainly grass), and individuals have permanent access to free water in ponds, and access to refuges and basking sites made of pile of rocks. Both ground and avian predation are avoided with plastic polyvinyl chloride (PVC) walls and with nets. As in the wild, main pressures on survival in the enclosures are thus social interactions, energy depletion, or parasites (Massot et al. 1992; Le Galliard et al. 2004). Hibernation usually lasts from October to February–March. Mating season usually begins upon emergence of females in March–April and lasts between 2 to 4 weeks. Gestation lasts usually 2 months but is highly sensitive to climatic conditions. This species is lecithotrophic implying that most of the energy supply is provided with yolk produced during vitellogenesis (Bleu et al. 2012). During pregnancy, there is no evidence of nutrient sharing between the mother and the embryo, and maternal effort is therefore oriented through mineral exchanges such as water and oxygen. Litter size ranges from 1 up to 12 juveniles (5 on average). There is no post-natal parental care. Body growth is continuous though life and decelerates with age. In our study populations, age at maturation varies from sometimes 1 (yearlings) to most of the time 2 years old (adults). Yearling individuals are distinguishable from older lizards by smaller body size, the majority not being sexually mature and not being as brightly colored as older lizards.

Study sample and rearing conditions

In May 2015, animals were captured in outdoor enclosures and identified by their unique "toe clip." The day of capture, each lizard was measured for snout-vent length (SVL ± 1 mm) and body mass (BM ± 1 mg). Lizards were then sorted in individual terraria with sterilized peat soil and kept under natural day-night light and temperature conditions in a temperature-controlled room (16 h of night at 16 °C and 8 h of day at 23 °C). Each terrarium provided lizards with a thermal gradient (23 to 35 °C) 6 h a day (09:00 to 12:00 and 14:00 to 17:00) with a light bulb (25 W) at one end of the terrarium. Animals were fed with house crickets three times per week (Acheta domestica, 300 ± 10 mg for yearlings; 400 ± 20 mg for more than 2-year-old adults). In this study, we used 171 pregnant females (151 adults, mean \pm SD: SVL = 65.8 \pm 3.6 mm, BM = 5.83 ± 1.05 g; 20 yearlings: SVL = $56.6 \pm$ 1.8 mm, BM = 3.36 ± 0.31 g). Water was available ad libitum in a bowl (petri dish) and misted three times a day (09:00, 13:00, 17:00).

Experimental manipulation of the pre-natal environment

We randomly selected and exposed 87 females (more than 2year-old adults, n = 77, yearlings, n = 10) to a mild water restriction period between 01 and 15 June 2015 whatever their stage in gestation. In this case, the water bowl was removed and water was misted only once in the morning (09:00). At the same time, 84 control females (more than 2-year-old adults, n = 74, yearlings, n = 10) were maintained with the ad libitum water conditions described above. These treatments mimic natural conditions with permanent (e.g., bog habitat) or periodic access to water (e.g., dry meadows where morning dew is the only available source of free water; Dupoué et al. 2017). Previous studies showed that water restriction induced significant changes in body mass, plasma osmolality (i.e., an index of body hydration), and activity time of lizards (Lorenzon et al. 1999; Dupoué et al. 2017).

Experimental manipulation of the post-natal environment

Females gave birth from 23 June to 14 July 2015; juveniles were sexed counting their ventral scales (Lecomte et al. 1992). The day of parturition, two to four juveniles per litter (equal sex ratio whenever possible) were randomly selected from each litter (171 litters, 422 juveniles; 204 females and 218 males). Exploration behavior was measured the day following birth (i.e., juvenile stage) using neutral arena tests. The experiment was done in a temperature-controlled room (25 °C) during activity time of the day (09:00 to 17:00). Each behavioral test was performed during 30 min in a neutral arena (25 \times 15×15 cm) heated from above with a lamp (40 W) and illuminated with white light. During the first 10 min, the individuals had time to recover from handling stress and to habituate to the arena. We used the last 20 min to record the immediate behavioral responses. The acclimation time allowed for lizards to initiate exploration and the recording time typically corresponds to the time period where individuals normally explore the space of the neutral arena (Le Galliard et al. 2015). We randomly assigned each arena either to a dry or a wet treatment (i.e., post-natal environment). The substratum (fine sand) was dried before each trial at 150 °C during 2 h in a stove. In the dry treatment, dry sand was used, and in the wet treatment, the sand was sprayed homogeneously with water (ca. 30 mL) immediately before the trial to saturate the sand with water. Sand was changed, and terraria were cleaned between each trial in order to avoid the presence of conspecifics' odor. Each lizard was tested four times (i.e., four trials) with two repeats per soil moisture treatment. Each lizard was tested with a pseudo-random sequence to ensure that treatment was not confounded with trial number. Each trial was recorded with a webcam at a high frequency (five images/second). Behavioral experiment lasted from 24 June to 15 July and included 1474 analyzable videos out of 1688 records. Nonusable records had corrupted files or a camera with an inappropriate angle of view.

After experiments, each litter was randomly released in 1 out of the 24 outdoor enclosures between early July and early August. In May 2016, we recaptured all individuals that survived over the year with a high success of capture (Le Galliard et al. 2005). The day of capture, individuals were identified with their toe clipping, measured, and weighed. We calculated annual growth rate (mm.day⁻¹) by dividing the SVL increment from birth to recapture by the total number of days between birth and recapture. Due to a lower survival rate than the average, inter-annual portrait (survival rate of 14% similar to mean value reported in Dupoué et al. 2017 but lower than the ca. 30%

inter-annual mean), we recaptured only 26 males and 34 females from the sample of individuals tested at juvenile stage. We also recaptured individuals that had not been tested but that were from the same litters (37 males, 25 females). Yearlings were kept for 3 days in the laboratory without food to ensure that all individuals were in a similar post-absorptive state. We then recorded their behavior in neutral arenas $(44.5 \times 24 \times 26 \text{ cm})$ larger than those used earlier in life to scale with the body size increase (ca. 100 mL). Setups also differed in the location of the heat source: at juvenile stage, heat lamps were above the corner of the box, while they were centered above the box at the yearling stage. Except for arena size and light placement, the exact same neutral arena protocol as the one used for juveniles was used for yearlings. Experiment lasted 8 days between 19 May and 30 May, and 479 usable video records were obtained.

Behavioral data

To minimize observer bias, blinded computer-based methods were used when all behavioral data were processed. We extracted the spatial coordinates of each lizard inside the neutral arena and the spatial location of edges of the neutral arena with a threshold detection procedure in ImageJ (version 1.49, https://imagej.nih.gov). We used the movement path records to calculate several relevant measures of the exploration and movement behavior. First, we calculated the distance between two locations from two consecutive images (movement step length). This variable followed a clear bimodal distribution with the lowest mode corresponding to motionless lizards. We thus extracted the second mode of the distribution from each record using the Rmixmod package in R (Lebret et al. 2015) and calculated the value of the 99 percentile of the distribution with the lowest mean. When distance values were below this distribution, lizards were considered inactive; otherwise, a positive step length (i.e., distance between two locations of two consecutive images) was recorded. We calculated the relative time spent active (T), the total distance walked by the lizard (D), the number of activity bouts (i.e., number of sequences of consecutive images with positive step length, N_{bouts}), the average distance walked in a bout (D_{bouts}), the average distance walked between two images in a bout (M_{bouts}), and the mean standard deviation of distances walked between two images in a boot (homogeneity of displacements; SD_{bouts}). Second, we added a buffer zone along the walls of the neutral arena (corresponding approximately to a lizard length) in order to measure the behavior of the lizard when they moved along or close to the wall. Movements towards or away from the walls are called "thigmotaxis" behavior and have been used as an index of "anxiety" in rodents (Treit and Fundytus 1988; Simon et al. 1994) and in lizards (Maximino et al. 2014). Thigmotaxis is not frequently measured in behavioral ecology studies, but Maximino et al. (2014) observed that this behavior is inhibited by anxiolytics in a reptile, when exploratory behaviors were enhanced, indicating that exploration and thigmotaxis are indeed distinct behavioral traits. We calculated the time spent by the lizard in the buffer zone along the walls (T_{walls}), the total distance walked out of the buffer zone $(D_{1 \text{ walls}})$, the mean distance to the walls (D_{2walls}), and the standard deviation of the distance to the walls (SD_{walls}). Third, we subdivided the neutral arena into 24 equal squares, to estimate the total area visited by the lizards including the proportion of squares visited at least once by the individual (τ_{explo}) and the standard deviation of the number of locations per square or heterogeneity of the exploration (SD_{explo}) .

Statistical analyses

All statistical analyses were carried out in R version 3.3.0 (R Core Team 2016). Prior to analyses, the distribution of each behavioral trait was transformed using a Box-Cox power function with "AID" package to meet the requirements of the Gaussian distribution (Asar et al. 2017). The identification of suites of correlated behavioral traits (i.e., behavioral syndromes) was explored using principal component analyses (PCAs) on the centered and scaled individual values of each behavioral trait with the "ade4" package (Dray and Dufour 2007). The number of components in the analysis was chosen using the broken stick method (Legendre and Legendre 1998), and we identified the contribution of each variable to each dimension of the PCA by calculating the inertia. A contribution was significant when it was higher to the average contribution (Legendre and Legendre 1998). In the subsequent analyses, each PC axis was considered as a behavioral syndrome; individual PC scores were extracted, and individuals were ranked according to them. This analysis was conducted on both years separately. We also repeated this analysis with a data set of juveniles and yearling records in order to study repeatability of behavioral syndromes during early life (see below).

Possibly due to handling and novelty stress, some individuals were motionless during the whole duration of a test leading to a bimodal distribution of behavioral traits related to mobility and exploration (see below). We calculated the proportion of these records and excluded these immobile individuals from our analyses of mobility and exploration behavior. To do so, we fitted the best bimodal distribution to the data based on loglikelihood maximization and deleted observations that had more than 50% chance of being part of the motionless mode of the distribution.

Question 1: are behavioral syndromes repeatable within and between life stages?

We calculated a repeatability index of the PC scores using a linear mixed model (LMM, Verbeke and Molenberghs 2009; package nlme) including the soil moisture treatment (i.e., dry or wet) as fixed effect and individual identity as a random effect. Repeatability R was calculated from the variance components following model results (see Eq. 1). The model was run with separate data from each year with a random individual variability around the intercept. Significance of the intra-individual variance was tested with a loglikelihood ratio test (LRT). We further tested the correlations of behaviors between the 2 years. Since PC axes were similarly structured across years (see below), we merged data from the 2 years from recaptured individuals and calculated a mean PC score for each year for each individual. We then tested the correlation between the individual scores of the 2 years.

$$R = \frac{Var_{Interindividual}}{Var_{Interindividual} + Var_{Residual}}$$
(1)

Question 2: are behavioral syndromes a result from pre-natal and post-natal interactions?

We tested the effects of the maternal treatment (i.e., water restriction or control), soil moisture treatment (i.e., dry or wet), trial number (i.e., trials 1, 2, 3 to 4), and their interactions on behaviors. We built LMMs with each PC score as a response variable, and lizard identity, mother identity, and the enclosure of origin of the mother as random factors to account for repeated measures on the same individual, nonindependence among siblings, and differences among enclosures, respectively. The full model included a three-way interaction between trial, maternal treatment, and the soil moisture treatment. When testing behaviors at birth, we also included the standardized time since first laying, a measure of the developmental stage of offspring in utero during the maternal treatment, in two-way interactions with both treatments. We first checked for the Gaussian distribution and homoscedasticity of the residuals of the full model. We then selected a minimum adequate model using backward elimination of the non-significant terms based on LRT. We first tested the significance of random effects and then removed non-significant fixed effects.

Question 3: do juvenile exploration behaviors influence their fitness?

For each individual record at birth, we calculated the mean individual PC score from the four measurements, and the

plasticity of each PC score (difference between mean score in the wet and the dry conditions). We tested the effects of mean and plasticity in PC scores on annual survival using generalized linear mixed models (GLMM, package lme4) with a binomial error. We also tested the effects of mean and plasticity in PC scores in interaction with maternal treatment on survival. We tested the effects on annual growth rate using LMMs. Mother and enclosure of release identity were treated as additive random effects.

Results

Repeatability of behavioral syndromes

In both years, the broken stick method selected only two significant PC axes (Table 1). Variables associated with locomotor activity (distance walked by the lizard, number of movement boots, distance per activity bout, average distance between each images of bouts, standard deviation of the distance, distance walked along the walls) and exploration

Table 1 Eigenvalues and relative contributions of each measured behavioral variables (in percentage) to the variance of each PCA axes. A contribution is considered significant when it is higher than the mean (i.e., 8.33%). The sign of the contribution indicates if it positively (no sign) or negatively (-) correlates with the PC axis. Behavioral traits are total distance walked by the lizard (D), relative time spent active (T), number of activity bouts (N_{bouts}), average distance walked in a bout (D_{bouts}), average distance walked between two images (M_{bouts}), mean standard deviation of distances walked between two images in a bout (Sd_{bouts}), time spent by the lizard in the buffer zone along the wall (T_{walls}) , total distance walked in the buffer zone (D_{1walls}) , mean distance to the wall when the lizard is in movement (D_{2walls}), standard deviation of the distance to the walls or heterogeneity of the distance to the walls (SD_{walls}), proportion of squares visited at least once by the individual (τ_{explo}) , and standard deviation of the number of locations in the square visited or heterogeneity of the exploration (SDexplo)

	2015		2016		Both			
	PC1	PC2	PC1	PC2	PC1	PC2		
Eigenvalues	52.5	19.4	49.8	26.1	57.7	19.7		
D	13.46	4.58	16.10	-0.12	11.82	6.10		
Т	9.69	3.70	13.29	0.12	10.20	2.52		
N _{bouts}	1.63	3.23	3.27	0.04	5.70	0.001		
M _{bouts}	11.84	3.92	14.10	-0.66	10.88	4.32		
SD _{bouts}	12.34	3.59	13.55	-0.15	10.27	6.35		
D _{bouts}	11.91	2.88	11.67	-0.33	7.95	8.47		
Twalls	3.50	-21.56	-1.11	-25.35	3.84	-23.25		
D _{1walls}	10.77	-7.72	14.94	1.52	10.93	-0.58		
D _{2walls}	0.20	-31.15	-0.86	-28.07	1.23	-29.04		
SD _{walls}	6.79	-14.98	-0.29	-28.10	7.58	-15.81		
τ_{explo}	9.25	-2.69	4.35	-12.90	9.05	2.61		
SD _{explo}	- 8.62	0.001	-6.50	2.65	- 10.55	0.95		

(proportion of time spent walking, proportion of the space visited) were positively correlated with the first axis (PC1). Lizards with positive values of the PC1 score were also characterized by a lower heterogeneity of exploration suggesting that all the space was visited equally. Thus, PC1 provided a score of activity and exploration in the neutral arena, which we subsequently describe as the exploration score. On the other hand, three behavioral variables related to movement relative to the walls of the neutral arena were correlated with the second axis (PC2, Table 1) with negative correlations with mean distance to the walls and time spent out the buffer zone and negative correlation with standard deviation of the distance to the walls. Thus, PC2 provided a score of thigmotaxis.

Short-term repeatability was higher in 2016 in yearlings (PC1: R = 64.9% and PC2: 39.8%) than in 2015 (PC1: 27.4, PC2: 17.9), but was significant in each life stage (juveniles, PC1: LRT = 104.33, p < 0.001; PC2: LRT = 56.6, p < 0.001; yearlings, PC1: LRT = 228.34, p< 0.001; PC2: LRT = 86.4, p < 0.001). In addition, PC scores at the yearling stage were not significantly correlated with PC scores at the juvenile stage (PC1: Pearson's correlation coefficient $R^2 = 0.08$, t = 0.6, df = 58, p = 0.53; PC2: $R^2 = -0.08$, t = -0.6, df = 58, p = 0.55).

Maternal effects on behavioral syndromes

Exploration score was best fitted with a bimodal distribution in 2015 and 2016 (Supplementary Fig. 1): 147 observations in 2015 and 17 in 2016 were considered as immobility records and were removed from the dataset. The probability of being motionless was significantly higher during the first trial and the lowest during the third trial (LRT = 98.74, p < 0.001) and in wet environments (LRT = 26.59, p < 0.001) at the juvenile stage. In yearlings, this probability was also greater explained by the trial number (LRT = 27.15, p < 0.001), but was not related to neutral arena wetness (LRT = 1.95, p = 0.16). In yearlings, the maternal treatment did not affect the probability of being motionless (juveniles: LRT = 26.59, p < 0.001; yearlings: LRT = 0.10, p = 0.75).

The best models for each PC score had no significant effects of the mother enclosure identity (PC1 in 2015: LRT = 1.09, p = 0.30; PC2 in 2015: LRT = 0.06, p = 0.81; PC1 in 2016: LRT = 1.60, p = 0.21; PC2 in 2016: LRT = 1.46, p = 0.23; see details on the best models in Tables 2 and 3). Exploration score was not influenced by an interaction between soil moisture treatment and maternal treatment (juveniles: LRT = 0.96, p = 0.37; yearlings: LRT = 0.69, p = 0.41; Fig. 1). In juveniles, exploration was higher for individuals from water-restricted mothers than for controls when mothers laid early; on the contrary, individuals from latest clutches from water-restricted mother had a lower exploration rate than controls (Table 2). Maternal treatment significantly enhanced thigmotaxis score at the juvenile stage (Table 3, Fig. 1c), but

Table 2 Best statistical models describing significant effects ofmaternal water restriction treatment, soil moisture treatment, and trialnumber on exploration behaviors. Minimum adequate models wereselected by backward elimination of non-significant terms (see maintext). Description of the effect of each numeric variable and of eachlevel of the factors in the best models selected for exploration rate in

juveniles and yearlings. These effects are presented as the difference to the reference value called intercept (trial 1; wet soil moisture conditions; control mother). In italics are effects that are significantly different from 0. The standard deviation explained by the random effects in each model are also displayed

	ndf	ddf	F statistics	p value		Estimate	SE	DF	t value	p value
Exploration score at the juvenile stage (N = 1327)										
Trial number	3	900	33.00	< 0.0001	Intercept	-0.93	0.18	900	- 5.15	< 0.0001
					Trial 2	0.98	0.21	900	4.67	< 0.0001
					Trial 3	1.72	0.17	900	9.86	< 0.0001
					Trial 4	1.17	0.21	900	5.66	< 0.0001
Soil moisture treatment	1	900	30.87	< 0.0001	Dry	1.16	0.21	900	5.56	< 0.0001
Trial number \times soil moisture treatment	3	900	4.74	0.0028	Trial 2: dry	-0.32	0.34	900	-0.96	0.3374
					Trial 3: dry	-0.82	0.23	900	-3.55	0.0004
					Trial 4: dry	-0.12	0.34	900	-0.37	0.7127
Scaled time since first laying	1	246	0.24	0.62		0.06	0.12	246	0.49	0.6219
Scaled time since first laying × soil moisture treatment	1	900	7.98	0.0048	Dry	0.22	0.08	900	2.83	0.0048
Maternal water restriction treatment	1	169	0.76	0.39	Stress	0.13	0.15	169	0.87	0.3856
Scaled time since first laying × mat. water restric. treat.	1	246	7.04	0.0085	Stress	-0.41	0.15	246	-2.65	0.0085
Random effects						Intercept		Residual		
					Mother	0.7274				
					Juvenile in mother	0.6293		1.3829		
Exploration score at the yearling stage $(N = 463)$										
Soil moisture treatment	1	340	21.29	< 0.0001	Intercept	0.10	0.19	340	0.54	0.5855
					Dry	-0.60	0.12	340	-5.03	< 0.0001
Random effects						Intercept		Resid	lual	
					Mother	0.7594				
					Juvenile in mother	0.6247		1.388	38	

had no effect at the yearling stage (LRT = 1.12, p = 0.29) and did not interact with soil moisture treatment (juveniles: LRT = 0.51, p = 0.47; yearlings: LRT = 1.29, p = 0.26; Fig. 1).

Exploration score was higher in dry than in wet soil moisture treatment in both life stages (Fig. 1a, b). In addition, in juveniles, this effect was stronger in individuals from latest clutches (Table 2). Thigmotaxis score was higher in dry than in wet soil moisture conditions at the juvenile stage (Fig. 1c), but this difference was reversed at the yearling stage (Fig. 1d). Exploration score of juveniles increased on average with trial number until the last trial, where scores were slightly lower than in trial 3, and the difference between wet and dry soil moisture conditions was maximum in the first and last trial (Fig. 2). Similar effects of trial number were found for mean exploration score of yearlings and mean thigmotaxis score of juveniles (Fig. 2c). On the contrary, thigmotaxis score of yearlings decreased with trial number (Fig. 2d).

Relationship between behavioral traits and fitness components

Annual survival was not influenced by the mean exploration score at birth (LRT = 0.59, p = 0.44), plasticity in exploration score (LRT = 0.72, p = 0.40), mean thigmotaxis score (LRT = 1.01, p = 0.32), nor by plasticity of thigmotaxis score (LRT = 0.84, p = 0.36). Survival was not influenced by the interaction of the maternal treatment with the mean exploration score at birth (LRT = 0.47, p = 0.49), plasticity in exploration score (LRT = 0.41, p = 0.52), mean thigmotaxis score (LRT = 0.55, p = 0.46), nor by plasticity of thigmotaxis score (LRT = 1.00, p = 0.32). Similarly, annual growth rate was not impacted by the mean exploration score at birth ($F_{1,7} = 0.38$, p = 0.56), plasticity of exploration score ($F_{1,7} = 0.14$, p = 0.72), mean thigmotaxis score ($F_{1,7} = 0.04$, p = 0.84), nor by plasticity of thigmotaxis score ($F_{1,7} = 2.66$, p = 0.15). **Table 3** Best statistical models describing significant effects ofmaternal water restriction treatment, soil moisture treatment, and trialnumber on thigmotaxis behaviors. Minimum adequate models wereselected by backward elimination of non-significant terms (see maintext). Description of the effect of each numeric variable and of eachlevel of the factors in the best models selected for thigmotaxis rate in

juveniles and yearlings. These effects are presented as the difference to the reference value called Intercept (trial 1; wet soil moisture conditions; control mother). In italics are effects that are significantly different from 0. The standard deviation explained by the random effects in each model are also displayed

	ndf	ddf	F statistics	p value		Estimate	SE	DF	t value	p value
Thigmotaxis score at the juvenile stage $(N = 1474)$									i	
Trial number	3	1048	5.89	0.0006	Intercept	-0.89	0.17	901	-5.38	< 0.0001
					Trial 2	0.98	0.21	901	4.62	< 0.0001
					Trial 3	1.75	0.18	901	10.00	< 0.0001
					Trial 4	1.16	0.21	901	5.58	< 0.0001
Maternal water restriction treatment	1	169	4.99	0.0267	Stressed mother	0.23	0.10	169	2.23	0.0267
Soil moisture treatment	1	1048	54.64	< 0.0001	Dry conditions	0.53	0.07	1048	7.39	< 0.0001
						Intercept		Residu	ıal	
					Mother	0.3334				
					Juvenile in mother	0.5355		1.3483	5	
Thigmotaxis score at the yearling stage $(N=480)$										
Trial number	3	354	4.33	0.0052	Intercept	0.56	0.18	354	3.16	0.0017
					Trial 2	-0.26	0.17	354	-1.50	0.1351
					Trial 3	-0.36	0.17	354	-2.06	0.0398
					Trial 4	-0.63	0.18	354	-3.58	0.0004
Soil moisture treatment	1	354	16.02	0.0001	Dry conditions	-0.49	0.12	354	-4.00	0.0001
						Intercept		Residu	ıal	
					Mother	0.5484				
					Yearling in mother	0.9678		1.3427	1	

Discussion

We investigated the age-dependent effects of pre-natal and post-natal hydric conditions on space use behavior of offspring. We used the neutral arena test, which provides a simple and standard procedure to score a variety of exploration, locomotor, and space use behaviors in ground-dwelling vertebrates such as rodents or lizards (Réale et al. 2007; Gould et al. 2009; Le Galliard et al. 2013). In addition, by using the neutral arena test under two conditions representing extremes from a substratum wetness gradient, we were able to score not only the mean lizard behavior but also the behavioral flexibility of individuals. We acknowledge that this simple test protocol did not allow to quantify the boldness of individuals and their habitat choice behaviors that could be tested in further studies with more complex arenas including shelters and micro-habitats. Our results show that two behavioral scores related to exploration and thigmotaxis (a proxy of anxiety) could be extracted from movement paths of lizards at both life stages.

Exploration behavior relates to the ability to gather spatial information in a novel or changing environment and has been measured by metrics of locomotor activity, spatial sampling, and information gathering similar to

ours in recent studies with lizards (e.g., Rodríguez-Prieto et al. 2011). As expected, exploration behavior was enhanced in the dry soil moisture treatment at the juvenile and yearling life stages in line with previous studies of natal dispersal behavior in the same species (Massot et al. 2002). Individuals did not differ consistently in the behavioral sensitivity of exploration to soil moisture treatment. We also found that individuals from latest clutches increased their exploration behavior in response to the dry soil treatment with more amplitude than individuals from early clutches. This indicates differences in the behavioral flexibility of offspring associated with differences in laying dates of females, most probably because of some unknown maternal effect. In general, dry environments may be challenging for lizards on the very short term given the physiological constraints on water balance (e.g., higher rate of water losses; Summers and Norman 1988; Dupoué et al. 2015a). A more pronounced locomotor activity and spatial sampling may therefore indicate more active exploration to evade such conditions and search for water. Such enhanced behavioral exploration could be crucial for survival, especially in juvenile lizards that disperse quickly in the month following birth. Individuals



Fig. 1 Behavioral syndrome estimated for **a** PC1 in 2015 (juveniles), **b** PC1 in 2016 (yearlings), **c** PC2 in 2015 (juveniles), and **d** PC2 in 2016 (yearlings) for each treatments. Soil moisture treatments are differentiated with colors: dark gray symbols are for wet soil moisture conditions, and light gray symbols are for dry soil moisture conditions. Maternal treatments are differentiated with the shape of the symbols: circles are for control mothers, and squares are for water restricted mothers. Light

gray dashed violin plots represent raw data distributions. Error bars are for 95% confidence intervals. Stars represent *p* values. n.s. non-significant; 0.05 < * < 0.01; 0.01 < ** < 0.001; *** < 0.001. Upper *p* values are comparing soil moisture treatments; lower *p* values are comparing maternal treatments. Significance of the effects and their estimates are presented in Tables 2 and 3. R library "ggplot2" and Inkscape 0.91 were used to create the artwork

showing more exploration behavior have indeed a higher probability to disperse in this species (Massot et al. 2002; Cote and Clobert 2007). Yet, we did not observe any influence of neither behavioral traits nor behavioral plasticity of juveniles on their annual survival. As all enclosures were similar in density and predation was prevented, food availability, parasitism, meteorological conditions, and social interactions were the main selection pressures in this study (Le Galliard et al. 2015). The absence of natural selection on exploration behavior could therefore be the consequence of a low predation, although it was also reported in previous studies including some allowing for predation mortality (Le Galliard et al. 2013, 2015).

In a previous study, Lorenzon et al. (1999) demonstrated that a chronic water restriction decreased locomotor activity in yearling common lizards in contrary to what we found in this study simulating an acute change in hydric conditions. This highlights that chronic exposure to water stress might trigger specific behavioral adjustments. Individuals may initiate a flight response by increasing exploration behavior on the short

 \diamond

3

n.s

 \diamond

4

4



Fig. 2 Behavioral syndrome estimated for a PC1 in 2015 (juveniles), b PC1 in 2016 (yearlings), c PC2 in 2015 (juveniles), and d PC2 in 2016 (yearlings) for each trials. In a, soil moisture treatments are differentiated by the color of the symbols: light gray squares for dry soil moisture conditions, and dark gray circles for wet soil moisture conditions. In b, c, and d, as soil moisture treatments had no effect, gray diamonds are for trials whatever the treatment. Light gray dashed violin plots represent raw

data distributions. Error bars are for 95% confidence intervals. Stars represent p values comparing each estimate to the estimate in trial 1. n.s. non-significant; 0.05 < * < 0.01; 0.01 < ** < 0.001; *** < 0.001. Significance of the effects and their estimates are presented in Tables 2 and 3. R library "ggplot2" and Inkscape 0.91 were used to create the artwork

3

term (Greenberg 2002). Whenever the exposure to dry conditions becomes more chronic, individuals might then decrease exploration and switch to energy-saving mechanisms that would reduce the rate of water loss and limit the risks of dehydration. Future studies would benefit from examining the behavioral responses to short- or long-term exposure to changes in hydric conditions.

Thigmotaxis relates to anxiety in neutral arena tests with laboratory rodents (Simon et al. 1994). Therefore, the proportion of time spent near the walls relative to the time spent elsewhere can be a good proxy of this behavior (Maximino et al. 2014) and was best summarized by PC2 score in this study. We observed a higher thigmotaxis score in dry soil moisture conditions in juveniles, consistently with the hypothesis that dry conditions are stressful and initiate a flight response. For example, lizards exposed to an acute physiological stress display more scratching behaviors (de Fraipont et al. 2000). However in yearlings, the thigmotaxis score was lower in dry soil moisture conditions despite the fact that lizards were more active and exploratory on average at this life stage. We expected that behaviors would be less sensitive to habitat humidity in yearlings because older animals should be more resistant to desiccation (due to their bigger size) and yearling common lizards disperse much less than juveniles (Newman and Dunham 1994; Massot et al. 2002; Meylan et al. 2002), but we did not predict an opposite response. In addition, we do not know how stressful are dry soil moisture conditions since we did not measure any indicator of physiological stress (hormones or body mass change for example). It is however unlikely that our short-term behavioral test treatment caused a significant water imbalance and subsequent stress responses because average water fluxes imply significant physiological responses within hours not minutes. Instead, we suggest that the immediate behavioral responses of lizards allow them to anticipate the future physiological costs of dehydration in the dry soil treatment.

Pre-natal exposure to water restriction increased the thigmotaxis behavior of juveniles independently from soil moisture treatment, but environmental conditions during gestation did not influence the behavior of yearlings. These results highlight that a stress-related behavior (thigmotaxis) scored at birth was positively influenced by the level of hydric stress experienced by females during pregnancy. Water restriction can result in higher physiological stress levels as shown by higher plasma corticosterone levels (e.g., Dauphin-Villemant and Xavier 1987; Dupoué et al. 2016), and this was especially true for mothers during gestation in this study (Dupoué et al. 2017). In turn, increased maternal physiological stress could explain the higher thigmotaxis of their offspring (de Fraipont et al. 2000). For example, prenatal stress can potentially reduce glucocorticoid receptors in offspring with subsequent rise in the levels of anxiety (Maccari et al. 1995, 2003). These results thus indicate that water restriction during gestation enhanced maternal physiological stress, which led to a higher behavioral anxiety in their offspring. This change in thigmotaxis in response to pre-natal water stress suggests that stressinduced natal dispersal phenotypes might be mediated by thigmotaxis behavior in this species. We know very little about the role of anxiety in natal dispersal in animals (Cote et al. 2010), including in the common lizard, and the numerous studies using neutral arena tests have rarely quantified thigmotaxis in the same way than we did. Some earlier studies suggested that physiological and behavioral stress can influence natal dispersal and juvenile fitness (Belthoff and Dufty 1998; Meylan et al. 2002), but we did not measure natal dispersal in this study, and thigmotaxis was unrelated to annual survival. Annual survival was on average lower in juveniles from water-restricted females (Dupoué et al. 2017), but this was not explained by the higher behavioral anxiety of these juveniles at birth. Further studies should test if individuals showing more thigmotaxis at birth disperse more or further distances.

On average, pre-natal exposure to water restriction did not influence exploration behavior at birth, but contrasted effects were found depending on the laying date and therefore developmental stage of juveniles. In early clutches, i.e., in juveniles that experienced water restriction late in their development (i.e., 3 weeks to 10 days before birth), water restriction enhanced exploration. In late clutches, i.e., from juveniles that experienced water restriction during the middle of their development (i.e., 4-6 weeks before birth), water restriction decreased exploration. These stage-dependent effects of water restriction on offspring exploration are difficult to explain a posteriori. Water demands for embryos increase late in gestation in viviparous reptiles (Dupoué et al. 2015b; Lourdais et al. 2015). Water restriction at that stage may induce stronger physiological stress responses in gravid females and signals deteriorating environmental conditions that promote natal dispersal and associated increase in exploration behaviors. The relationship between exploration behavior and natal dispersal has indeed been well documented in numerous animal taxa (Dingemanse et al. 2003; Clobert et al. 2009; Debeffe et al. 2013). Instead, water restriction in the middle of development may impair juvenile quality and thus reduce their ability to explore. Previous studies argue that negative maternal effects on juvenile survival and morphology are stronger when they occur earlier in development (Heath et al. 1999; Lindström 1999). Future studies should test the links between water restriction, exploration behaviors, and life history traits with a protocol imposing the treatment explicitly at the beginning or at the end of gestation.

The inter- and intra-individual variability of behavioral traits we observed in this study also gave us further insights. First of all, correlation structure between all behavioral variables, allowing us to calculate exploration and thigmotaxis scores, remained qualitatively the same between the 2 years of study. Yet, no correlation was found between the juvenile and yearling behavioral scores, which is consistent with some other studies that observed strong intra-individual changes in personality across development (e.g., Bell and Stamps 2004). In addition, behaviors were less repeatable at birth than at the yearling stage: exploration rate and thigmotaxis rate showed a repeatability score high enough to be subjectively considered as personalities only at the yearling stage (Réale et al. 2007). A higher repeatability of exploration early in life was expected because natal dispersal mostly occurs few months after birth in this species and natal dispersal has been correlated to consistent behavioral traits such as exploration rate in several other species (Dingemanse et al. 2003). On the other hand, Hoset et al. (2011) observed a positive correlation between exploration behavior and dispersal status in a small rodent; still, there was a low repeatability of this behavioral trait like in our study. One possibility is that the exploration personality type was not well developed in the first days of life when individuals were more sensitive to environmental changes,

including handling and procedural stress. It is now accepted that personalities are not always stable in time during ontogeny or when the environment changes importantly (Stamps and Groodhuis 2010) and that exposure to environmental conditions early in life can drive the emergence of alternative personality types (e.g., Bell and Sih 2007).

Our results indicate that the behavioral responses of organisms to water availability involved independent pre-natal and post-natal effects of hydric conditions on behavioral traits associated with exploration and anxiety. In general, dry conditions and water restriction elicited flight responses (i.e., higher exploration and anxiety), especially in juveniles and late in development, but prenatal and post-natal environmental conditions influenced differently behavioral traits related to exploration and anxiety. Exploration and anxiety were sensitive to acute changes in hydric conditions in both life stages. Maternal effects influenced average anxiety of juveniles but were not significant at the yearling stage suggesting vanishing long-term effects of the maternal environment on offspring behavior. These results indicate that different behavioral items are involved in plastic responses to water restriction during embryogenesis and fetal development than in behavioral flexibility. The behavioral response of an organism to a changing environment is thus likely to be a combination of independent behavioral items shaped differently by stage-specific environmental cues. We suggest that the life history strategy linking together these behavioral changes in the European common lizard is natal dispersal. In order to elucidate better this complexity, additional behavioral traits and the natal dispersal phenotype should be tested, in particular the link between natal dispersal, anxiety syndromes, and environmental stress.

Acknowledgments We are really thankful to the anonymous reviewers who helped us improve the manuscript. We warmly thank Simon Agostini who helped in capturing the lizards and managing outside enclosures and Coralie Delme who helped in doing the experiments and breeding the animals.

Funding information This work was supported by funds from Centre National de la Recherche Scientifique (CNRS) and Agence Nationale de la Recherche grant (ANR-13-JSV7-0011-01).

Compliance with ethical standards

Conflict of interest The authors declare that they have no conflict of interest.

Ethical approval All applicable international, national, and/or institutional guidelines for the care and use of animals were followed. Animal care and behavioral protocols were performed in accordance with laws on animal experimentation in France and Europe, especially EU Directive 2010/63/EU, and were approved by national ethics committee for animal experimentation under file number APAFIS#5108-2016040811272391. Animal care and breeding was performed by authorized personnel under permit DTTP-2008-449 issued to J-FLG. Experiments were conducted under agreement A77-341-1 delivered by the Préfecture de Seine-et-Marne. Animals were captured and manipulated under authorization 2007-198-005 delivered by the Préfecture de Seine-et-Marne.

References

- Agrawal AA, Laforsch C, Tollrian R (1999) Transgenerational induction of defences in animals and plants. Nature 401(6748):60–63. https:// doi.org/10.1038/43425
- Asar Ö, Ilk O, Dag O (2017) Estimating Box-Cox power transformation parameter via goodness-of-fit tests. Commun Stat 46:91–105.
- Bell AM, Stamps JA (2004) Development of behavioural differences between individuals and populations of sticklebacks, *Gasterosteus* aculeatus. Anim Behav 68(6):1339–1348. https://doi.org/10.1016/j. anbehav.2004.05.007
- Bell AM, Sih A (2007) Exposure to predation generates personality in threespined sticklebacks (*Gasterosteus aculeatus*). Ecol Lett 10(9): 828–834. https://doi.org/10.1111/j.1461-0248.2007.01081.x
- Belthoff JR, Dufty AMJ Jr (1998) Corticosterone, body condition and locomotor activity: a model for dispersal in screech-owls. Anim Behav 55:405–415, 2, DOI: https://doi.org/10.1006/anbe.1997.0625
- Bleu J, Massot M, Haussy C, Meylan S (2012) Experimental litter size reduction reveals costs of gestation and delayed effects on offspring in a viviparous lizard. Proc R Soc Lond B 279(1728):489–498. https://doi.org/10.1098/rspb.2011.0966
- Chown SL, Gaston KJ (1999) Exploring links between physiology and ecology at macro-scales: the role of respiratory metabolism in insects. Biol Rev 74(1):87–120. https://doi.org/10.1017/ S000632319800526X
- Chown SL, Sørensen JG, Terblanche JS (2011) Water loss in insects: an environmental change perspective. J Insect Physiol 57(8):1070– 1084. https://doi.org/10.1016/j.jinsphys.2011.05.004
- Clobert J, Galliard L, Cote J, Meylan S, Massot M (2009) Informed dispersal, heterogeneity in animal dispersal syndromes and the dynamics of spatially structured populations. Ecol Lett 12:197–209
- Cote J, Clobert J (2007) Social personalities influence natal dispersal in a lizard. Proc R Soc Lond B 274(1608):383–390. https://doi.org/10. 1098/rspb.2006.3734
- Cote J, Clobert J, Brodin T, Fogarty S, Sih A (2010) Personalitydependent dispersal: characterization, ontogeny and consequences for spatially structured populations. Philos T Roy Soc B 365(1560): 4065–4076. https://doi.org/10.1098/rstb.2010.0176
- Dall SRX, Houston AI, McNamara JM (2004) The behavioural ecology of personality: consistent individual differences from an adaptive perspective. Ecol Lett 7(8):734–739. https://doi.org/10.1111/j. 1461-0248.2004.00618.x
- Dauphin-Villemant C, Xavier F (1987) Nychthemeral variations of plasma corticosteroids in captive female *Lacerta vivipara* Jacquin: influence of stress and reproductive state. Gen Comp Endocrinol 67(3):292–302. https://doi.org/10.1016/0016-6480(87)90183-3
- Davis JR, DeNardo DF (2009) Water supplementation affects the behavioral and physiological ecology of Gila monsters (*Heloderma suspectum*) in the Sonoran Desert. Physiol Biochem Zool 82(6): 739–748. https://doi.org/10.1086/605933
- de Fraipont M, Clobert J, John H, Meylan S (2000) Increased pre-natal maternal corticosterone promotes philopatry of offspring in common lizards *Lacerta vivipara*. J Anim Ecol 69(3):404–413. https:// doi.org/10.1046/j.1365-2656.2000.00405.x
- Debeffe L, Morellet N, Cargnelutti B, Lourtet B, Coulon A, Gaillard JM, Bon R, Hewison AJM (2013) Exploration as a key component of natal dispersal: dispersers explore more than philopatric individuals

in roe deer. Anim Behav 86(1):143-151. https://doi.org/10.1016/j. anbehav.2013.05.005

- Dingemanse NJ, Both C, van Noordwijk AJ, Rutten AL, Drent PJ (2003) Natal dispersal and personalities in great tits (*Parus major*). Proc R Soc Lond B 270(1516):741–747. https://doi.org/10.1098/rspb.2002.2300
- Dingemanse NJ, Dochtermann NA (2013) Quantifying individual variation in behaviour: mixed-effect modelling approaches. J Anim Ecol 82(1):39–54. https://doi.org/10.1111/1365-2656.12013
- Dingemanse NJ, Kazem AJ, Réale D, Wright J (2010) Behavioural reaction norms: animal personality meets individual plasticity. Trends Ecol Evol 25:81–89
- Dray S, Dufour AB (2007) The ade4 package: implementing the duality diagram for ecologists. J Stat Softw 22:1–20
- Dupoué A, Angelier F, Brischoux F, DeNardo DF, Trouvé C, Parenteau C, Lourdais O (2016) Water deprivation increases maternal corticosterone levels and enhances offspring growth in the snake *Vipera aspis*. J Exp Biol 219(5):658–667. https://doi.org/10.1242/jeb.132639
- Dupoué A, Brischoux F, Angelier F, DeNardo DF, Wright CD, Lourdais O (2015b) Intergenerational trade-off for water may induce a mother– offspring conflict in favour of embryos in a viviparous snake. Funct Ecol 29(3):414–422. https://doi.org/10.1111/1365-2435.12349
- Dupoué A, Galliard L, Josserand R, DeNardo DF, Decencière B, Agostini S, Haussy C, Meylan S (2017) Water restriction causes an intergenerational trade-off and delayed mother-offspring conflict in a viviparous lizard. Funct Ecol (published online, doi:https://doi.org/10. 1111/1365-2435.13009
- Dupoué A, Stahlschmidt ZR, Michaud B, Lourdais O (2015a) Physiological state influences evaporative water loss and microclimate preference in the snake *Vipera aspis*. Physiol Behav 144:82– 89. https://doi.org/10.1016/j.physbeh.2015.02.042
- Forstmeier W, Coltman DW, Birkhead TR (2004) Maternal effects influence the sexual behavior of sons and daughters in the zebra finch. Evolution 58(11):2574–2583. https://doi.org/10.1111/j.0014-3820. 2004.tb00885.x
- Gould TD, Dao DT, Kovacsics CE (2009) The open field test. In: Gould TD (ed) Mood and anxiety related phenotypes in mice. Humana Press, New York, pp 1–20. https://doi.org/10.1007/978-1-60761-303-9 1
- Greenberg N (2002) Ethological aspects of stress in a model lizard, *Anolis carolinensis*. Integr Comp Biol 42(3):526–540. https://doi.org/10. 1093/icb/42.3.526
- Heath DD, Fox CW, Heath JW (1999) Maternal effects on offspring size: variation through early development of Chinook salmon. Evolution 53(5):1605–1611. https://doi.org/10.1111/j.1558-5646.1999. tb05424.x
- Hertz PE, Arce-Hernandez A, Ramirez-Vazquez J, Tirado-Rivera W, Vazquez-Vives L (1979) Geographical variation of heat sensitivity and water loss rates in the tropical lizard, *Anolis gundlachi*. Comp Biochem Physiol A 62:947–953
- Hoset KS, Ferchaud A-L, Dufour F, Mersch D, Cote J, Le Galliard J-F (2011) Natal dispersal correlates with behavioral traits that are not consistent across early life stages. Behav Ecol 22(1):176–183. https://doi.org/10.1093/beheco/arq188
- Kearney MR, Simpson SJ, Raubenheimer D, Kooijman SA (2013) Balancing heat, water and nutrients under environmental change: a thermodynamic niche framework. Funct Ecol 27(4):950–966. https://doi.org/10.1111/1365-2435.12020
- Köhler A, Sadowska J, Olszewska J, Trzeciak P, Berger-Tal O, Tracy CR (2011) Staying warm or moist? Operative temperature and thermal preferences of common frogs (*Rana temporaria*), and effects on locomotion. Herpetol J 21:17–26
- Le Galliard J-F, Clobert J, Ferriere R (2004) Physical performance and Darwinian fitness in lizards. Nature 432:502–505
- Le Galliard J-F, Ferriere R, Clobert J (2005) Juvenile growth and survival under dietary restriction: are males and females equal? Oikos 111: 368–376

- Le Galliard J-F, Paquet M, Cisel M, Montes-Poloni L (2013) Personality and the pace-of-life syndrome: variation and selection on exploration, metabolism and locomotor performances. Funct Ecol 27(1): 136–144. https://doi.org/10.1111/1365-2435.12017
- Le Galliard J-F, Paquet M, Mugabo M (2015) An experimental test of density-dependent selection on temperament traits of activity, boldness and sociability. J Evol Biol 28:1144–1155
- Lebret R, Iovleff S, Langrognet F, Biernacki C, Celeux G, Govaert G (2015) Rmixmod: the r package of the model-based unsupervised, supervised and semi-supervised classification mixmod library. J Stat Softw 67:241–270
- Lecomte J, Clobert J, Massot M (1992) Sex identification in juveniles of Lacerta vivipara. Amphibia-Reptilia 13:21–25
- Legendre P, Legendre L (1998) Numerical ecology, 2nd, English edn. Elsevier Science, Amsterdam
- Lima SL, Dill LM (1990) Behavioral decisions made under the risk of predation: a review and prospectus. Can J Zool 68(4):619–640. https://doi.org/10.1139/z90-092
- Lindström J (1999) Early development and fitness in birds and mammals. Trends Ecol Evol 14:343–348
- Lorenzon P, Clobert J, Oppliger A, John-Alder H (1999) Effect of water constraint on growth rate, activity and body temperature of yearling common lizard (*Lacerta vivipara*). Oecologia 118(4):423–430. https://doi.org/10.1007/s004420050744
- Lorenzon P, Clobert J, Massot M (2001) The contribution of phenotypic plasticity to adaptation in Lacerta vivipara. Evolution 55(2):392– 404. https://doi.org/10.1111/j.0014-3820.2001.tb01302.x
- Lourdais O, Lorioux S, Dupoué A, Wright C, DeNardo DF (2015) Embryonic water uptake during pregnancy is stage-and fecundity-dependent in the snake *Vipera aspis*. Comp Biochem Physiol A 189:102–106
- Maccari S, Darnaudery M, Morley-Fletcher S, Zuena AR, Cinque C, Van Reeth O (2003) Prenatal stress and long-term consequences: implications of glucocorticoid hormones. Neurosci Biobehav Rev 27(1-2):119–127. https://doi.org/10.1016/S0149-7634(03)00014-9
- Maccari S, Piazza PV, Kabbai M, Barbazanges A, Simon H, Le Moal M (1995) Feedback induced by prenatal stress. J Neurosci 15(1 Pt 1): 110–116
- Massot M, Clobert J, Lorenzon P, Rossi J-M (2002) Condition-dependent dispersal and ontogeny of the dispersal behaviour: an experimental approach. J Anim Ecol 71(2):253–261. https://doi.org/10.1046/j. 1365-2656.2002.00592.x
- Massot M, Clobert J, Pilorge T, Lecomte J, Barbault R (1992) Density dependence in the common lizard: demographic consequences of a density manipulation. Ecology 73:1742–1756
- Maximino C, Carvalho CM, Morato S (2014) Discrimination of anxietyversus panic-like behavior in the wall lizard *Tropidurus oreadicus*. Psychol Neurosci 7(2):227–231. https://doi.org/10.3922/j.psns. 2014.020
- McCormick MI (2006) Mothers matter: crowding leads to stressed mothers and smaller offspring in marine fish. Ecology 87(5):1104– 1109.
- Meylan S, Belliure J, Clobert J, de Fraipont M (2002) Stress and body condition as prenatal and postnatal determinants of dispersal in the common lizard (*Lacerta vivipara*). Horm Behav 42(3):319–326. https://doi.org/10.1006/hbeh.2002.1827
- Mousseau TA, Fox CW (1998) The adaptive significance of maternal effects. Trends Ecol Evol 13:403–407
- Newman RA, Dunham AE (1994) Size at metamorphosis and water loss in a desert anuran (*Scaphiopus couchii*). Copeia 1994(2):372–381. https://doi.org/10.2307/1446984
- Pintor AF, Schwarzkopf L, Krockenberger AK (2016) Hydroregulation in a tropical dry-skinned ectotherm. Oecologia 182:925–931
- Pough FH, Taigen TL, Stewart MM, Brussard PF (1983) Behavioral modification of evaporative water loss by a Puerto Rican frog. Ecology 64(2):244–252. https://doi.org/10.2307/1937072

- R Core Team (2016) R: language and environment for statistical computing. R Foundation for Statistical Computing, Vienna http://www.Rproject.org
- Rautenberg W, May B, Arabin G (1980) Behavioral and autonomic temperature regulation in competition with food intake and water balance of pigeons. Pflugers Arch 384:253–260
- Réale D, Reader SM, Sol D, McDougall PT, Dingemanse NJ (2007) Integrating animal temperament within ecology and evolution. Biol Rev 82(2):291–318. https://doi.org/10.1111/j.1469-185X. 2007.00010.x
- Rodríguez-Prieto I, Martín J, Fernández-Juricic E (2011) Individual variation in behavioural plasticity: direct and indirect effects of boldness, exploration and sociability on habituation to predators in lizards. Proc R Soc Lond B 278:266–273
- Rozen-Rechels D, van Beest FM, Richard E, Uzal A, Medill SA, McLoughlin PD (2015) Density-dependent, central-place foraging in a grazing herbivore: competition and tradeoffs in time allocation near water. Oikos 124(9):1142–1150. https://doi.org/10.1111/oik.02207
- Seebacher F, Alford RA (2002) Shelter microhabitats determine body temperature and dehydration rates of a terrestrial amphibian (*Bufo marinus*). J Herpetol 36(1):69–75.
- Shochat E, Lerman SB, Katti M, Lewis DB, Brown AEJS (2004) Linking optimal foraging behavior to bird community structure in an urbandesert landscape: field experiments with artificial food patches. Am Nat 164(2):232–243. https://doi.org/10.1086/422222

- Sih A, Bell A, Johnson JC (2004) Behavioral syndromes: an ecological and evolutionary overview. Trends Ecol Evol 19:372–378
- Simon P, Dupuis R, Costentin J (1994) Thigmotaxis as an index of anxiety in mice. Influence of dopaminergic transmissions. Behav Brain Res 61:59–64
- Stamps J, Groothuis TG (2010) The development of animal personality: relevance, concepts and perspectives. Biol Rev 85(2):301–325. https://doi.org/10.1111/j.1469-185X.2009.00103.x
- Summers CH, Norman MF (1988) Chronic low humidity-stress in the lizard Anolis carolinensis: changes in diurnal corticosterone rhythms. J Exp Zool 247(3):271–278. https://doi.org/10.1002/jez. 1402470311
- Treit D, Fundytus M (1988) Thigmotaxis as a test for anxiolytic activity in rats. Pharmacol Biochem Behav 31(4):959–962. https://doi.org/10. 1016/0091-3057(88)90413-3
- Verbeke G, Molenberghs G (2009) Linear mixed models for longitudinal data. Springer, New York
- Versteegh MA, Helm B, Dingemanse NJ, Tieleman BI (2008) Repeatability and individual correlates of basal metabolic rate and total evaporative water loss in birds: a case study in European stonechats. Comp Biochem Physiol A 150(4):452–457. https://doi.org/ 10.1016/j.cbpa.2008.05.006
- Woods HA, Smith JN (2010) Universal model for water costs of gas exchange by animals and plants. P Natl Acad Sci USA 107(18): 8469–8474. https://doi.org/10.1073/pnas.0905185107